Author's response to reviews

Title: A Mediation Model of Mindfulness and Decentering: Sequential psychological constructs or one and the same?

Authors:

Judith Gecht (j.gecht@psych.rwth-aachen.de)
Ramona Kessel (rkessel@ukaachen.de)
Thomas Forkmann (tforkmann@ukaachen.de)
Siegfried Gauggel (sgauggel@ukaachen.de)
Barbara Drueke (bdrueke@ukaachen.de)
Anne Scherer (ascherer@ukaachen.de)
Verena Mainz (vmainz@ukaachen.de)

Version: 3  Date: 2 April 2014

Author's response to reviews:

Dear Editor,

the changes made concern the inclusion of line numbering and the authors' contributions section as well as the Acknowledgements section. Furthermore, the reference style has been changed conform to the journal's requirements.

Yours sincerely,

Judith Gecht